Dear Editor,

First, we thank you for your comments and support to improve the manuscript. We have addressed all comments and uploaded the clean version of the revised manuscript. Please find below our rebuttal to the comments.

Best,

Shimels

************************************************
Editor Comments:
1. While assessing your manuscript we found several instances where the text displayed overlap with other previously published works, in particular:

https://doi.org/10.1371/journal.pone.0209220
Overlap with this previously published work was found mainly in:
the last 6 lines of page 13 &,
The first five lines of page fourteen.
While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. We would therefore be grateful if you could reformulate the aforementioned section of your paper to resolve the overlap between your manuscript and other sources. Please ensure that, where relevant, these sources are also referenced as appropriate.

If this study uses methodology from a previously published work, please provide a summarizing statement in the methodology together with a citation to the original paper.

Response: The comment is right. We now have changed (reformulated) the statements that coincided with our other work. Thank you for the comment. Page 13, last 6 lines, to page 14, first 3 lines.

2. In the Ethics approval and consent to participate subsection of the Declarations please clarify if the caregivers of participants provided their consent in written or verbal form. If consent was provided verbally, please include reasoning as to why this method was selected over written consent. Please also include a statement as to whether your ethics committee explicitly approved the method of verbal consent.

Response: It was an informed written consent and this has now been clearly indicated.

3. Please change the section heading which currently reads as “Introduction” so that it instead reads as “Background.”

Response: Corrected.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: done as advised. Thank you for the support and cooperation.